Author’s response to reviews

Title: Neonatal hearing screening: modelling cost and effectiveness of hospital- and community-based screening

Authors:

Eva Grill (eva.grill@med.uni-muenchen.de)
Kai Uus (kai.uus@man.ac.uk)
Franz Hessel (franz.hessel@uni-essen.de)
Linda Davies (Linda.Davies@man.ac.uk)
Rod S Taylor (R.S.Taylor@bham.ac.uk)
Juergen Wasem (juergen.wasem@uni-essen.de)
John Bamford (John.Bamford@man.ac.uk)

Version: 3  Date: 22 February 2006

Author’s response to reviews: see over
Dear Alison,

Thank you for the directions and help for the manuscript formatting. Enclosed please find the revised version of the manuscript “Neonatal hearing screening: modelling cost and effectiveness of hospital- and community-based screening” which we would like to submit to BMC Health Services Research.

We made the following changes:

**Title page** – Statement about equal contribution was omitted.

**Competing interests** – We indicate ‘The authors declare that they have no competing interests’.

**Highlighting/Tracking** - removed

**Figures** – Figures were closely cropped.

**Tables** – Each table has a title

Best wishes,

Dr. Eva Grill, MPH